Author's response to reviews

Title: Criterion validation of two submaximal aerobic fitness tests, the self-monitoring Fox-walk test and the Astrand cycle test in people with rheumatoid arthritis

Authors:

Birgitta Nordgren (birgitta.nordgren@ki.se)
Cecilia Fridén (cecilia.friden@ki.se)
Eva Jansson (eva.jansson@ki.se)
Ted Österlund (ted.osterlund@karolinska.se)
Wilhelmus Grooten (wim.grooten@ki.se)
Christina H Opava (christina.opava@ki.se)
Anette Rickenlund (anette.rickenlund@karolinska.se)

Version: 2
Date: 14 May 2014

Author's response to reviews:

The manuscript was considered to be too methodological and restricted in its scope to be of sufficient interest to the readership of Arthritis Research & Therapy and no revisions of the manuscript was expected to be done yet.